Massachusetts Institute of Technology Handout 0 6.042J/18.062J: Mathematics for Computer Science January 28, 1994 Professors Charles E. Leiserson and Tom Leighton

Information Sheet

This is the first term that 6.042J/18.062J has been offered, and consequently, it has a restricted enrollment of 40 students. Far more preregistered, however, so this information will be used to help determine who will be allowed to take it. Some of the information you provide will be made into a handout and distributed to the class.

Name Department and year Phone number Email address

Please list all Course VI subjects you have taken or are currently taking (mark the latter with an asterisk).

Please list all Course XVIII subjects you have taken or are currently taking (mark the latter with an asterisk).

Each student must attend a one-hour tutorial on Monday. Please circle the hours you prefer and cross out those you cannot attend.

10 11 12 1 2 3 4

Recitations will be held on Fridays. All three recitations were originally scheduled for 3:00 p.m.. Two of them have been rescheduled, however. One will meet at 1:00 p.m. in 26-210, one at 2:00 p.m. in 26-210, and the third remains at 3:00 p.m. in 34-303. Please circle the hours you prefer and cross out those you cannot attend.

1 2 3